Author's response to reviews

Title: Effects of Carissa opaca fruits extracts on oxidative pulmonary damages and fibrosis in rats

Authors:

sumaira sahreen (sumairasahreen@yahoo.com)
M R Khan (mrkhangau@yahoo.com)
Rahmat Khan (rahmatqu81@yahoo.com)

Version: 6 Date: 9 January 2014

Author's response to reviews: see over
To

The Editor,

BMC Complementary and Alternative Medicine

Dear Sir,

We are submitting a copyedited research paper entitled “Effects of Carissa opaca fruits extracts on oxidative pulmonary damages and fibrosis in rats” in your esteemed journal. There is not any conflict. All authors are agreed to submit this research paper in your esteemed journal. This paper has not been submitted or published anywhere.

With regards

Dr. Rahmat Ali Khan
Assistant professor
Department of Biotechnology, UST Bannu
Email: rahmatgul_81@yahoo.com

Editorial comments:

- Please ensure that the ORCiD identifier of Rahmat Khan is included in your Authors Contributions section.
  Ans: Added

- Edit the language.
  Ans: Edited from Endaz